Title: Effect of intra-antral application of antifungal agent in the treatment of recurrent maxillary sinusitis

Authors:

Adekunle D Dunmade (dumkun@yahoo.com)
Olushola A Afolabi (droaafolabi@yahoo.com)
Biodun S Alabi (alabibs@yahoo.com)
Segun Segun-Busari (segunsegunbusari@yahoo.com)
Olubisi A Koledoye (olubisikoledoye@yahoo.com)

Version: 2 Date: 23 February 2012

Author's response to reviews: see over
The Editor,

Journal of Medical Case Report,

Dear Sir,

**RE- EFFECT OF INTRA-ANTRAL APPLICATION OF ANTIFUNGAL AGENT IN THE TREATMENT OF RECURRENT MAXILLARY FUNGAL RHINOSINUSITIS.**

We wish to submit the above titled manuscript to your highly esteemed journal. We the author of the above titled article wish to state that the article has not been submitted elsewhere and it is not been considered for publication elsewhere.

This case report will add to the scientific knowledge by improving ways of managing a recurrent fungal maxillary sinusitis using a topical antifungal agent. This we applied intra-antrally in a patient who had recurrence of the maxillary sinusitis after debulking and commencing systemic antifungal agent which is an unusual presentation in sinusitis. This will also help to reduce the toxicity as the systemic drug can be safely replace with topical agents. If published the authors shall transfer the copyright to your journal sir.

Also attached is the copyright form sir.

Thanking you in anticipation

Yours Faithfully,

Signed

**Dr O.A. Afolabi**

(Correspondence author)